Author's response to reviews

Title: The need to modify physical activity messages to better speak to older African American women: a qualitative study

Authors:

Emerson Sebastiao (esebast2@illinois.edu)
Andiara Schwingel (andiara@illinois.edu)
Wojtek Chodzko-Zajko (wojtek@illinois.edu)

Version: 2
Date: 1 April 2015

Author's response to reviews:

Revisions made:
(1) Title page has been removed;
(2) A new structured abstract has been created;
(3) Line number were added throughout the manuscript;
(4) Tables were places after the reference.